Author's response to reviews

Title: Eradication of intractable malignant ascites by abdominocentesis, reinfusion of concentrated ascites, and adoptive immunotherapy with dendritic cells and activated killer cells in a recurrent lung cancer patient: a case report

Authors:

Hideki Kimura (hkimura@chiba-cc.jp)
Toshihiko Iizasa (tiizasa@chiba-cc.jp)
Aki Ishikawa (aishikawa@chiba-cc.jp)
Mitsuru Yoshino (myoshino@chiba-cc.jp)
Masato Shingyouji (mshingyouji@chiba-cc.jp)
Masaki Kimura (mkimura@chiba-cc.jp)
Tetushi Hirata (thirata@chiba-cc.jp)
Akiko Odaka (aodaka@chiba-cc.jp)
Keiko Matsubayasi (kmatsuba@chiba-cc.jp)

Version: 11 Date: 29 June 2008

Author's response to reviews:

Dear editorial board of JMCR
Thank you for your kind advices. We have added the sentence to express the exploratory nature of this report.
p-10,l-4
Although it is very difficult to apply these treatment to other lung cancer patients because recurrence only in abdominal cavity as malignant ascites is very rare, further study is necessary to analyze the efficacy and mechanisms of this exploratory treatment.
Sincerely yours
Hideki Kimura M.D. Ph.D.
Director Chiba cancer Center, Japan